Author's response to reviews

Title: Impact of insufficient drug efficacy of antiparkinson agents on patient's quality of life: A cross-sectional study

Authors:

Jun Tsugawa (tsujun1@msn.com)
Rieko Onozawa (riekoo@fukuoka-u.ac.jp)
Jiro Fukae (jfukae@fukuoka-u.ac.jp)
Takayasu Mishima (mishima1006@fukuoka-u.ac.jp)
Shinsuke Fujioka (shinsuke@cis.fukuoka-u.ac.jp)
Yoshio Tsuboi (tsuboi@cis.fukuoka-u.ac.jp)

Version: 2 Date: 28 August 2014

Author's response to reviews:

Timothy Shipley, MD,
Executive Editor, BMC Neurology
BioMed Central

Dear Dr. Timothy Shipley

We would like to submit the revised version of our manuscript, entitled “Impact of insufficient drug of agents on patient’s quality of life: A cross-sectional study” for consideration for publication in BMC Neurology as a Research article.

We have added line and page numbers in our manuscript. We hope that our corrections and changes to the manuscript are satisfactory and we look forward to hearing from you.

Sincerely,

Jun Tsugawa, MD
Department of Neurology, Fukuoka University

Yoshio Tsuboi, MD (Correspondence)
Professor of Neurology, Fukuoka University
7-45-1 Nanakuma, Jonan-ku, Fukuoka 814-0180, Japan
tsuboi@cis.fukuoka-u.ac.jp